Author's response to reviews

Title: Management of patients with biliary sphincter of Oddi disorder without sphincter of Oddi manometry

Authors:

Evangelos Kalaitzakis (evangelos.kalaitzakis@medicine.gu.se)
Tim Ambrose (tjwambrose@hotmail.com)
Jane Phillips-Hughes (jane.phillips-hughes@orh.nhs.uk)
Jane Collier (jane.collier@orh.nhs.uk)
Roger Chapman (roger.chapman@ndm.ox.ac.uk)

Version: 3 Date: 20 October 2010

Author's response to reviews:

Dear Sir or Madam,

the requested changes have been made in the manuscript (author contributions, highlighted areas, removal of vertical lines from the tables). The manuscript has been checked for typos.

Kind regards

Evangelos Kalaitzakis MD, PhD
Associate professor of medicine
Sahlgrenska Academy
University of Gothenburg
Gothenburg
Sweden